Author’s response to reviews

Title: Cognitive profile in patients with a first-ever lacunar infarct with and without silent lacunes: a comparative study.

Authors:

Lorena Blanco-Rojas (l.blanco.rojas@copc.cat)
Adria Arboix (aarboix@hscor.com)
David Canovas (dcanovas@tauli.cat)
Marta Grau-Olivares (martagrau76@hotmail.com)
Joan Carles Soliva Morera (jcsoliva@tauli.cat)
Olga Parra (oparra@ub.edu)

Version: 2 Date: 29 November 2012

Author’s response to reviews:

To the Executive Editor of BMC Neurology,

Dear Deesha Majithia:

Enclosed please find the changes made in the Methods section of our manuscript (page 4): “Prior to conducting the study, approval was obtained from Ethical Committee on Clinical Research of the hospital and complied with the Declaration of Helsinki. Written informed consent was obtained from all patients”.

We would like to thank you in advance for your attention to this paper.

Sincerely,

Adrià Arboix, MD, PhD
Cerebrovascular Division
Department of Neurology
Capio-Hospital Universitari del Sagrat Cor
Viladomat 288
E-08029 Barcelona
Spain
Tel.: +34 93 4948940, fax +34 93 4948906
E-mail: aarboix@hscor.com